\documentclass[12pt]{article}

\setlength{\textheight}{9in}
\setlength{\voffset}{-1in}
\setlength{\textwidth}{6in}
\setlength{\hoffset}{-0.5in}

\pagestyle{empty}
\setlength{\parindent}{0pt}

\begin{document}
\begin{center}
  \textbf{Math 325, Spring 2001}
\end{center}
Name:

\vfill


What is your major?

\vfill

Why are you taking Math 325?

\vfill

Have differential equations come up much in courses you have taken 
other than Math 228?  If so, in what courses?  What kind of equations?

\vfill 

\vfill

Do kind of graphing calculator do you own?

\vfill


What is your favorite book?

% \vfill
% What is your favorite food?

\end{document}

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 
